Author’s response to reviews

Title: Correlation and predictability of ocular aberrations and the visual outcome after quadrifocal intraocular lens implantation: a retrospective longitudinal study

Authors:

Chia Yi Lee (ao6u.3msn@hotmail.com)
Jing-Yang Huang (wchinyang@gmail.com)
Chi-Chin Sun (arvin.sun@msa.hinet.net)
Shun-Fa Yang (ysf@csmu.edu.tw)
Hung-Chi Chen (mr3756@cgmh.org.tw)
Hung-Yu Lin (anthonyhungyulin@hotmail.com)

Version: 4 Date: 02 Aug 2019

Author’s response to reviews:

Dear Editor Joann J. Kang and Jashan Abraham:

We are very glad to hear from BMC Ophthalmology again that a revision of our manuscript is warranted. We revised the manuscript according to the suggestions of the editor, and provided a list of the responses we made for all the comments which are located in the following sections. We hope that the responses can fulfill the requests and our manuscript will soon be accepted and published by BMC Ophthalmology.

Best wishes,

Chia-Yi Lee, MD
Department of Ophthalmology, Show Chwan Memorial Hospital
No.2, Ln. 530, Sec. 1, Zhongshan Rd., Changhua City, Changhua County 50093, Taiwan
Tel: 886-4-7256166
Fax: 886-4-7126911
E-mail: ao6u.3msn@hotmail.com
Editor Comments:
In order to help protect participant anonymity, please limit the identifiable information presented in Table 6. A good way to do this is to present the age of the participants as a range (e.g. 20's-30's) or removing the age (or sex) information.

We ask this, as it is BioMed Central policy to not publish more than 2 indirect identifiers (age, sex, profession, etc.) without explicit consent for publication from the participants.

Currently, the age and gender information in conjunction with other details, may compromise patient anonymity.

Response: Thanks for the suggestion, we present the age of the participants as a range and deleted the sex information in the new Table 6.